

## **Answering Reviewers**

### **REVIEWER #1**

We thank Reviewer #1 for the comments on our work. No issues requiring resolution have been indicated.

### **REVIEWER #2**

We thank Reviewer #2 for the comments on our work. No issues requiring resolution have been indicated.

### **EDITORIAL OFFICE'S COMMENTS**

1. We have obtained a language editing certificate using Grammarly Pro Version, provided by Department of Medical-Surgical Sciences and Translational Medicine, Sapienza University of Rome
2. We have modified the title in accordance with the provided indications.
3. We have modified Author list in accordance with the provided indications.
4. We have modified Authors and institution(s) in accordance with the provided indications.
5. Author contributions have been added in the text.
6. All references have been updated in accordance with the provided instructions (edit with Auto-Analyser at <https://www.f6publishing.com/Forms/main/ArticleReferenceTool.aspx>)
7. We have modified corresponding authors information in accordance with the provided indications.
8. We have provided a report for the plagiarism check.

### **COMPANY EDITOR-IN-CHIEF:**

We thank the company editor-in-chief for the comments on our work. No issues requiring resolution have been indicated.

9. A Conflict-of-interest statement has been added in the text.